% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P3.44}{23(3)}{2016}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf Recurrence relations for linear transformations\\ preserving the strong $q$-log-convexity}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address 

\author{Lily Li Liu\thanks{Supported by National Natural Science Foundation of China (Nos. 11471186, 11571150).}
\qquad\qquad
Ya-Nan Li\\ %\qquad  Forgotten Third Author\\
\small School of Mathematical Science\\[-0.8ex]
\small Qufu Normal University\\[-0.8ex] 
\small Qufu 273165, P.R. China\\
\small\tt liulily@mail.qfnu.edu.cn \  \   471009811@qq.com\\
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Feb 3, 2016}{Aug 26, 2016}{Sep 2, 2016}\\
\small Mathematics Subject Classifications: 05A20, 05A15, 11B37}

\begin{document}
\begin{thebibliography}{99}

\bibitem{mb97}
M. Benoumhani.
\newblock On some numbers related to Whitney numbers of Dowling lattices.
\newblock {\em Adv. in Appl. Math.}, 19:106--116, 1997.


\bibitem{mb99}
M. Benoumhani.
\newblock Log-concavity of Whitney numbers of Dowling lattices.
\newblock {\em Adv. in Appl. Math.},  22:186--189, 1999.

\bibitem{Bre89}
F. Brenti.
\newblock Unimodal, log-concave, and P\'olya frequency sequences in combinatorics.
\newblock {\em Mem. Amer. Math. Soc.}, 413, 1989.

\bibitem{Bro84}
A. Z. Broder.
\newblock The $r$-Stirling numbers.
\newblock {\em Discrete Math.}, 49:241--259, 1984.

\bibitem{cyc08}
W. Y. C. Chen, L. X. W. Wang, and A. L. B. Yang.
\newblock Recurrence relations for strongly $q$-log-convex polynomials.
\newblock {\em Canad. Math. Bull.}, 54:217--229, 2011.

\bibitem{cj12}
G.-S. Cheon, and J.-H. Jung.
\newblock The $r$-Whitney numbers of Dowling lattices.
 \newblock {\em Discrete Math.}, 15:2337--2348, 2012.
 
\bibitem{Com74}
L. Comtet.
\newblock {\em  Advanced Combinatorics}. Reidel, Dordrecht, 1974.


\bibitem{ta73}
T. A. Dowling.
\newblock A class of gemometric lattices bases on finite groups.
\newblock {\em J. Combin. Theory Ser. B}, 14:61--84,  1973;
\newblock {\em Erratum J. Combin. Theory Ser. B}, 15:211, 1973.

\bibitem{Kur72}
D. C. Kurtz.
\newblock A note on concavity properties of triangular arrays of numbers.
\newblock {\em J. Combin. Theory Ser. A}, 13:135--139, 1972.

\bibitem{LW-RZP}
L. L. Liu, and Y. Wang.
\newblock A unified approach to polynomial sequences with only real zeros.
\newblock {\em Adv. in Appl. Math.}, 38:542--560, 2007.

\bibitem{lwang07}
L. L. Liu, and Y. Wang.
\newblock On the log-convexity of combinatorial sequences.
\newblock {\em Adv. in Appl. Math.}, 39:453--476, 2007.


\bibitem{im11}
I. Mez\H{o}.
\newblock The $r$-Bell numbers.
\newblock {\em J. Integer Seq.}, 14: Article 11.1.1, 2011.

\bibitem{mr14}
M. Rahmani.
\newblock Some results on Whitney numbers of Dowling lattices.
\newblock {\em Arab J. Math. Sci.}, 20:11--27, 2014.



\bibitem{Sta89}
R. P. Stanley.
\newblock Log-concave and unimodal sequences in algebra, combinatorics, and geometry.
\newblock {\em Ann. New York Acad. Sci.}, 576:500--534, 1989.

\bibitem{WYjcta05}
Y. Wang, and Y.-N. Yeh.
\newblock Polynomials with real zeros and P\'olya frequency sequences.
\newblock {\em J. Combin. Theory Ser. A}, 109:63--74, 2005.

\bibitem{Zhu}
B.-X. Zhu.
\newblock Log-convexity and strong $q$-log-convexity for some triangular arrays.
\newblock {\em Adv. in Appl. Math.}, 50:595--606, 2013.

\bibitem{ZS15}
B.-X. Zhu, and H. Sun.
\newblock Linear transformations preserving the strong $q$-log-convexity of polynomials.
\newblock {\em Electron. J. Combin.}, 22:\#P3.27, 2015.





\end{thebibliography}


\maketitle

\begin{abstract}
Let $[T(n,k)]_{n,k\geqslant0}$ be a triangle of positive numbers
satisfying the three-term recurrence relation
$$
T(n,k)=(a_1n+a_2k+a_3)T(n-1,k)+(b_1n+b_2k+b_3)T(n-1,k-1).
$$
In this paper, 
we give a new sufficient condition for linear transformations
\begin{equation*}
Z_n(q)=\sum_{k=0}^{n}T(n,k)X_k(q)
\end{equation*}
that preserves the strong $q$-log-convexity of polynomials sequences.
As applications, 
we show linear transformations, given by matrices of the binomial coefficients, 
the Stirling numbers of the first kind and second kind, 
the Whitney numbers of the first kind and second kind,
 preserving the strong $q$-log-convexity in a unified manner.


  % keywords are optional
  \bigskip\noindent \textbf{Keywords:} Strong $q$-log-convexity; Linear transformation; Stirling number; Whitney number
\end{abstract}

\section{Introduction}

Let $(x_n)_{n\geqslant 0}$ be a sequence of nonnegative real numbers. 
It is said to be {\it log-convex}  {\it log-concave} resp.) if
$x_{n+1}x_m-x_nx_{m+1} \geqslant 0$ ($x_{n+1}x_m-x_nx_{m+1} \leqslant 0$ resp.) for $n\geqslant m\geqslant 0$.
Log-concave and log-convex sequences arise often in combinatorics.
We refer the reader to~\cite{Bre89,LW-RZP,Sta89,WYjcta05} for log-concavity and~\cite{lwang07,Zhu} for the log-convexity.

For two real polynomials $f(q)$ and $g(q)$, 
denote $f(q)\geqslant_q g(q)$ if the difference $f(q)-g(q)$ has only nonnegative coefficients. 
A real polynomial sequence
$(f_n(q))_{n\geqslant 0}$ is said to be \emph{$q$-log-convex} if
$$
f_{n+1}(q)f_{n-1}(q)\geqslant_q f_n^2(q)
$$
for $n\geqslant 1$, 
and is called \emph{strongly $q$-log-convex}
if
$$
f_{n+1}(q)f_m(q)\geqslant_qf_n(q)f_{m+1}(q)
$$
for $n\geqslant m\geqslant 0$. 
Clearly, the strong $q$-log-convexity of polynomials sequence implies the $q$-log-convexity. 
It is known that many famous polynomial sequences, 
such as the Bell polynomials, the Eulerian polynomials,
the Narayana polynomials and the Narayana polynomials of type $B$, 
are strongly $q$-log-convex~\cite{Zhu}. 
In this paper,
we study the strong $q$-log-convexity of polynomials sequences by using recurrence relations of their triangle.

Let $[T(n,k)]_{n,k\geqslant0}$ be a triangle of positive numbers satisfying the three-term recurrence relation
\begin{equation}\label{a-1}
T(n,k)=(a_1n+a_2k+a_3)T(n-1,k)+(b_1n+b_2k+b_3)T(n-1,k-1),
\end{equation}
with $T(n,k)=0$ unless $0\leqslant k\leqslant n$.  
It is natural to assume that $a_1n+a_2k+a_3\geqslant 0$ for $0\leqslant k<n$ and
$b_1n+b_2k+b_3\geqslant 0$ for $0<k\leqslant n$.
Note that the former is equivalent to $a_1\geqslant 0,a_1+a_2\geqslant 0,
a_1+a_3\geqslant 0$ and the latter is equivalent to $b_1\geqslant 0,b_1+b_2\geqslant 0,b_1+b_2+b_3\geqslant 0$. 
It is known that for each fixed $n$, 
the sequence $(T(n,k))_{0\leqslant k\leqslant n}$ is log-concave (Kurtz~\cite{Kur72}), 
and forms a P\'olya frequency sequence if $a_2b_1\geqslant a_1b_2$ and $a_2(b_1+b_2+b_3)\geqslant (a_1+a_3)b_2$ 
(Wang and Yeh~\cite[Corollary 3]{WYjcta05}). 
Let
$$T_n(q)=\sum_{k=0}^nT(n,k)q^k$$ 
be the row generating functions of $[T(n,k)]_{n,k\geqslant0}$. 
Liu and Wang~\cite{lwang07} obtained a sufficient condition for the $q$-log-convexity of the sequence $(T_n(q))_{n\geqslant 0}$.
Subsequently, 
Chen et al.~\cite{cyc08} discussed the strong $q$-log-convexity of the sequence $(T_n(q))_{n\geqslant 0}$.

We say that the linear transformation
\begin{equation}\label{lin-1}
Z_n(q)=\sum_{k=0}^{n}T(n,k)X_k(q)
\end{equation}
{\it preserves the strong $q$-log-convexity} if the strong $q$-log-convexity of the polynomials sequence
$(X_n(q))_{n\geqslant 0}$ implies the strong $q$-log-convexity of $(Z_n(q))_{n\geqslant 0}$. 
The corresponding triangle $[T(n,k)]_{n,k\geqslant0}$ is also said to be have the property of {\it preserving the strong $q$-log-convexity}.
Since the sequence $(q^k)_{k\geqslant0}$ is strongly $q$-log-convex,
the strong $q$-log-convexity of $(T_n(q))_{n\geqslant 0}$ can be deduced from the linear transformation~(\ref{lin-1}) preserving the strong $q$-log-convexity.

As pointed out by Brenti~\cite{Bre89},
 whenever we study a set of ``objects" in mathematics,
we look for linear transformations that preserve their structures.
There have been many results in combinatorics obtained using this idea.
For instance,
Wang and Yeh~\cite{WYjcta05} proved the log-concavity of combinatorial sequences by linear transformations.
Liu and Wang~\cite{lwang07} presented the log-convexity of  combinatorial sequences by linear transformations.
It is therefore natural to look for linear transformations that preserve the strong $q$-log-convexity.
Recently, Zhu and Sun~\cite{ZS15} obtained several linear transformations preserving the strong $q$-log-convexity.
In this paper,
we present some different linear transformations preserving the strong $q$-log-convexity.

In combinatorics,
many classical numbers satisfy the three-term recurrence.
For example,
the binomial coefficients satisfy
\begin{equation}\label{binom}\binom{n}{k}=\binom{n-1}{k}+\binom{n-1}{k-1}.\end{equation}
The Stirling numbers $S(n,k)$ of the second kind satisfy 
\begin{equation}\label{Stir}S(n,k)=S(n-1,k-1)+kS(n-1,k).\end{equation}
In this paper, using the three-term recurrence relation~(\ref{a-1}) of the triangle $[T(n,k)]_{n,k\geqslant0}$,
we give a new sufficient condition for linear transformations~(\ref{lin-1}) to preserve the strong $q$-log-convexity.


The rest of the paper is organized as follows. In
section 2, we obtain a sufficient condition for linear transformations~(\ref{lin-1}) to preserve the strong $q$-log-convexity.
As applications,  we show some linear transformations, given by matrices of the binomial coefficients,
the Stirling numbers of the second kind, the signless Stirling numbers of the first kind, the Whitney numbers of the second kind,
the signless Whitney numbers of the first kind and the Bessel numbers,
preserving the strong $q$-log-convexity in a unified manner.
Moreover, we obtain the strong $q$-log-convexity of the $r$-Dowling polynomials.




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Linear transformations preserving the strong $q$-log-convexity}

In this section,
we first recall the following result.

\begin{lemma}\label{l-2}\emph{(\cite{cyc08})}
Suppose that the triangle $[T(n,k)]_{n,k\geqslant0}$ of positive
numbers satisfies \emph{(\ref{a-1})} with $a_2,b_2\geqslant 0$. Then
for any $k\geqslant j\geqslant 0$, $n\geqslant m\geqslant 0$, we
have
\begin{equation}\label{a-2}
T(n,k)T(m,j)-T(n,j)T(m,k)\geqslant0.
\end{equation}
\end{lemma}

Then we present the main result of this paper.


\begin{theorem}\label{th-1}
Suppose that the triangle $[T(n,k)]_{n,k\geqslant0}$ of positive
numbers defined by \emph{(\ref{a-1})} with $a_2,b_2\geqslant 0$. Then
the linear transformation~(\ref{lin-1}) preserves the strong
$q$-log-convexity.
\end{theorem}

\begin{proof}
In order to prove the strong $q$-log-convexity of
$(Z_n(q))_{n\geqslant 0}$, it suffices to prove that
\begin{equation}\label{a-3}
Z_{n+1}(q)Z_m(q)-Z_n(q)Z_{m+1}(q)\geqslant_q 0
\end{equation}
for any $n\geqslant m\geqslant 0$.

Since the triangle $[T(n,k)]_{n,k\geqslant0}$ satisfies the
recurrence relation (\ref{a-1}), we have
\begin{eqnarray*}
&& Z_{n+1}(q)
= \sum_{k=0}^{n+1}T(n+1,k)X_k(q)\\
&=& \sum_{k=0}^{n+1}(a_1n+a_1+a_2k+a_3)T(n,k)X_k(q)+\sum_{k=0}^{n+1}(b_1n+b_1+b_2k+b_3)T(n,k)X_k(q)\\
&=&
\sum_{k=0}^{n}\left((a_1n+a_1+a_2k+a_3)X_k(q)+(b_1n+b_1+b_2k+b_2+b_3)X_{k+1}(q)\right)T(n-1,k).
\end{eqnarray*}

Substituting $Z_{n+1}(q)$ and $Z_{m+1}(q)$ into~(\ref{a-3}) yields

\begin{eqnarray*}
& & Z_{n+1}(q)Z_m(q)-Z_n(q)Z_{m+1}(q)\\
&=& \sum_{k=0}^{n}\sum_{j=0}^{m}(a_1n+a_1+a_2k+a_3)T(n,k)T(m,j)X_k(q)X_j(q)\\
& & +\sum_{k=0}^{n}\sum_{j=0}^m(b_1n+b_1+b_2k+b_2+b_3)T(n,k)T(m,j)X_{k+1}(q)X_j(q)\\
& & -\sum_{k=0}^{n}\sum_{j=0}^m(a_1m+a_1+a_2j+a_3)T(m,j)T(n,k)X_j(q)X_k(q)\\
& &
-\sum_{k=0}^{n}\sum_{j=0}^m(b_1m+b_1+b_2j+b_2+b_3)T(m,j)T(n,k)X_{j+1}(q)X_k(q)\\
&=&
\sum_{k=0}^{n}\sum_{j=0}^{m}[a_1(n-m)+a_2(k-j)]T(n,k)T(m,j)X_k(q)X_j(q)\\
& & +\sum_{k=0}^{n}\sum_{j=0}^{m}((b_1n+b_1+b_2k+b_2+b_3)X_{k+1}(q)X_j(q)\\
& & -(b_1m+b_1+b_2j+b_2+b_3)X_k(q)X_{j+1}(q))T(n,k)T(m,j)\\
&=& R_1(q)+R_2(q)+S_1(q)+S_2(q),
\end{eqnarray*}
where

$$R_1(q)=
\sum_{k=0}^{m}\sum_{j=0}^{m}(a_1(n-m)+a_2(k-j))T(n,k)T(m,j)X_k(q)X_j(q),$$
$$R_2(q)=
\sum_{k=m+1}^{n}\sum_{j=0}^{m}(a_1(n-m)+a_2(k-j))T(n,k)T(m,j)X_k(q)X_j(q),$$

\begin{eqnarray*}
S_1(q)&=& \sum_{k=0}^{m}\sum_{j=0}^{m}((b_1n+b_1+b_2k+b_2+b_3)X_{k+1}(q)X_j(q)\\
& &
-(b_1m+b_1+b_2j+b_2+b_3)X_k(q)X_{j+1}(q))T(n,k)T(m,j),\\
S_2(q)&=& \sum_{k=m+1}^{n}\sum_{j=0}^{m}((b_1n+b_1+b_2k+b_2+b_3)X_{k+1}(q)X_j(q)\\
& & -(b_1m+b_1+b_2j+b_2+b_3)X_k(q)X_{j+1}(q))T(n,k)T(m,j).
\end{eqnarray*}


Clearly,
$R_2(q),S_2(q)\geqslant_q 0$, for $k>m\geqslant j$. 
So to prove ~(\ref{a-3}), 
it suffices to prove $R_1(q),S_1(q)\geqslant_q 0$.


We first prove $R_1(q)\geqslant_q 0$.
Note that
\begin{eqnarray*}
R_1(q)&=& \sum_{k=0}^m\sum_{j=0}^ma_1(n-m)T(n,k)T(m,j)X_k(q)X_j(q)\\
& &
+\sum_{0\leqslant j<k\leqslant m}a_2(k-j)(T(n,k)T(m,j)-T(n,j)T(m,k))X_k(q)X_j(q).\\
\end{eqnarray*}
Thus by Lemma~\ref{l-2}, we have
$R_1(q)\geqslant_q 0.$


Then we prove $S_1(q)\geqslant_q 0$. 
Note that
\begin{eqnarray*}
S_1(q)&=& \sum_{0\leqslant j<k\leqslant m}((b_1n+b_1+b_2k+b_2+b_3)T(n,k)T(m,j)\\
&&-(b_1m+b_1+b_2k+b_2+b_3)T(n,j)T(m,k))X_{k+1}(q)X_j(q)\\
& & + \sum_{0\leqslant j<k\leqslant m}((b_1n+b_1+b_2j+b_2+b_3)T(n,j)T(m,k)\\
&&-(b_1m+b_1+b_2j+b_2+b_3)T(n,k)T(m,j))X_{j+1}(q)X_k(q)\\
& & +\sum_{k=0}^mb_1(n-m)X_{k+1}(q)X_k(q)T(n,k)T(m,k).
\end{eqnarray*}

Since $n\geqslant m\geqslant0$, we have
\begin{eqnarray*}
S_1(q)& \geqslant_q &\sum_{0\leqslant j<k\leqslant
m}((b_1n+b_1+b_2k+b_2+b_3)T(n,k)T(m,j)\\
&&-(b_1n+b_1+b_2k+b_2+b_3)T(n,j)T(m,k))X_{k+1}(q)X_j(q)\\
& & +\sum_{0\leqslant j<k\leqslant m}((b_1n+b_1+b_2j+b_2+b_3)T(n,j)T(m,k)\\
&&-(b_1n+b_1+b_2j+b_2+b_3)T(n,k)T(m,j))X_k(q)X_{j+1}(q)\\
& & +\sum_{k=0}^mb_1(n-m)X_{k+1}(q)X_k(q)T(n,k)T(m,k)\\
\end{eqnarray*}
\begin{eqnarray*}
&=& \sum_{0\leqslant j<k\leqslant m}(b_1n+b_1+b_2k+b_2+b_3)(T(n,k)T(m,j)-T(n,j)T(m,k))X_{k+1}(q)X_j(q)\\
& -& \sum_{0\leqslant j<k\leqslant m}(b_1n+b_1+b_2j+b_2+b_3)(T(n,k)T(m,j)-T(n,j)T(m,k))X_{k}(q)X_{j+1}(q)\\
& +& \sum_{k=0}^mb_1(n-m)X_{k+1}(q)X_k(q)T(n,k)T(m,k).
\end{eqnarray*}
On the other hand, for $k>j$, we have

\begin{eqnarray*}
S_1(q)  &\geqslant_q&
\sum_{0\leqslant j<k\leqslant m}(b_1n+b_1+b_2k+b_2+b_3)(T(n,k)T(m,j)-T(n,j)T(m,k))\\
& & (X_{k+1}(q)X_j(q)-X_{k}(q)X_{j+1}(q))\\
& & +\sum_{k=0}^mb_1(n-m)X_{k+1}(q)X_k(q)T(n,k)T(m,k).
\end{eqnarray*}


Thus we obtain $S_1(q)\geqslant_q 0$ from  Lemma~\ref{l-2} and the strong $q$-log-convexity of $(X_n(q))_{n\geqslant0}$.
The proof of the theorem is complete.
\end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Applications}
\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
In this section, 
as applications of Theorem~\ref{th-1},
we present some linear transformations preserving the strong $q$-log-convexity.

\subsection{Some classical linear transformations}
\hspace*{\parindent}

Let $s(n, k)$ and $S(n, k)$ be the Stirling numbers of the first kind and second kind respectively.
Denote by $c(n,k)=(-1)^{n+k}s(n,k)$ the signless Stirling numbers of the first kind, 
i.e., the number of permutations of $[n]$ with $k$ cycles.
It is known that the signless Stirling numbers of the first kind and the Stirling numbers of the second kind satisfy the recurrence relations

\begin{equation}\label{Stir-1}
c(n,k)=(n-1)c(n-1,k)+c(n-1,k-1),
\end{equation}
and
\begin{equation}\label{Stir}
 S(n,k)=S(n-1,k-1)+kS(n-1,k)
\end{equation}
respectively, with $c(n,0)=c(0,k)=S(n,0)=S(0,k)=0$, 
except for $c(0,0)=S(0,0)=1$ (see \cite{Com74} for instance).

Liu and Wang~\cite{lwang07} showed that the Stirling transformations of the first kind and second kind preserve the log-convexity.
Now we generalize it by 
showing that these two linear transformations also preserve the strong $q$-log-convexity by Theorem~\ref{th-1}.

\begin{corollary}\label{S-2}
The Stirling transformations of the first kind and second kind $$Z_n(q)=\sum_{k=0}^nc(n,k)X_k(q),\ \ \ \bar{Z}_n(q)=\sum_{k=0}^nS(n,k)X_k(q)$$ preserve the strong $q$-log-convexity respectively.
\end{corollary}

The Bell polynomials $B_n(q)=\sum_{k=0}^nS(n,k)q^k$ can be viewed as a $q$-analog of the Bell numbers and have many fascinating properties.
For example, the Bell polynomials $B_n(q)$ have only real zeros (see~\cite{WYjcta05} for instance).
Liu and Wang~\cite{lwang07} established that the Bell polynomials are $q$-log-convex.
Moreover, the Bell polynomials are strongly $q$-log-convex obtained by Chen et al.~\cite{cyc08} and Zhu~\cite{Zhu} respectively.
Now the strong $q$-log-convexity of $B_n(q)$ follows immediately from Corollary~\ref{S-2}.

\begin{corollary}[{\cite{cyc08,Zhu}}]
The Bell polynomials $B_n(q)$ form a strongly $q$-log-convex sequence.
\end{corollary}


The polynomial $F_n(q)=\sum_{k=0}^nk!S(n,k)q^k$ also admits a variety of combinatorial properties, 
such as the strong $q$-log-convexity~\cite{cyc08}.
Thus we have the following result from Corollary~\ref{S-2} and the strong $q$-log-convexity of $(k!q^k)_{k\geqslant0}$.

\begin{corollary}[{\cite{cyc08}}]
The polynomials $F_n(q)$ form a strongly $q$-log-convex sequence.
\end{corollary}



The Dowling lattice $Q_n(G)$ is a geometric lattice of rank $n$ over a finite group $G$ of order $m\geqslant 1$~\cite{ta73}. 
When $m=1$, that is, $G$ is the trivial group, 
$Q_n(G)$ is isomorphic to the lattice $\prod_{n+1}$ of partitions of an $(n+1)$-element set. 
So the Dowling lattices can be viewed as group-theoretic analogs of the partition lattices.
Let $w_{m}(n, k)$ and $W_{m}(n, k)$ be the $k$th Whitney numbers of the first kind and
second kind of the Dowling lattice respectively.
 Denote by $t_m(n,k)=(-1)^{n+k}w_{m}(n,k)$ the signless Whitney numbers of the first kind.
It is known that the signless Whitney numbers of the first kind and the Whitney numbers of the second kind satisfy the recurrence relations
$$t_m(n,k)=(1+m(n-1))t_m(n-1,k)+t_m(n-1,k-1)$$
and
$$W_m(n,k)=(1+mk)W_m(n-1,k)+W_m(n-1,k-1),\ \ \ n\geqslant k\geqslant 1$$
respectively, 
with $t_m(n,n)=t_m(n,0)=W_m(n,n)=W_m(n,0)=1$ for $n\geqslant0$ and $t_m(n,k)=W_m(n,k)=0$, except for $0\leqslant k\leqslant n$.
So we obtain the following result from Theorem~\ref{th-1}.


\begin{corollary}\label{W-2}
The Whitney transformations of the first kind and second kind $$Z_n(q)=\sum_{k=0}^nt_m(n,k)X_k(q),\ \ \  \bar{Z}_n(q)=\sum_{k=0}^nW_m(n,k)X_k(q)$$ preserve the strong $q$-log-convexity respectively for $m\geqslant 1$.
\end{corollary}


Benoumhani~\cite{mb97} introduced the Dowling polynomial $D_m(n,x)=\sum_{k=0}^nW_m(n,k)q^k$
and following two generalized polynomials
$F_{m,1}(n;q)=\sum_{k=0}^nk!W_m(n,k)m^kq^k,\  F_{m,2}(n;q)=\sum_{k=0}^nk!W_m(n,k)q^k.$
It is known that polynomials $D_m(n;q)$, $F_{m,1}(n;q)$ and $F_{m,2}(n;q)$ have only real zeros respectively (see~\cite{mb99,LW-RZP} for instance).
Chen et al.~\cite{cyc08} also prove the strong $q$-log-convexity of polynomials $D_m(n;q)$, $F_{m,1}(n;q)$ and $F_{m,2}(n;q)$.
It is easy to prove that sequences $(k!q^k)_{k\geqslant0}$ and $(k!m^kq^k)_{k\geqslant0}$ are strongly $q$-log-convex respectively.
Thus we have the following result from Corollary~\ref{W-2}.


\begin{corollary}[{\cite{cyc08}}]
The polynomials $D_m(n;q)$, $F_{m,1}(n;q)$ and $F_{m,2}(n;q)$ form strongly $q$-log-convex sequences respectively for $m\geqslant 1$.
\end{corollary}

The Bessel polynomials are defined by
$$y_n(q)=\sum_{k=0}^n\frac{(n+k)!}{(n-k)!k!}\left(\frac{q}{2}\right)^k,$$
and they have been extensively studied. 
Let $$Y(n,k)=\frac{(n+k)!}{(n-k)!k!}.$$
Chen et al.~\cite{cyc08} deduced that $[Y(n,k)]_{n,k\geqslant0}$ satisfies the following recurrence relation
$$Y(n,k)=Y(n-1,k)+(2n+2k-2)Y(n-1,k-1),\ \ \ n\geqslant k\geqslant 1.$$
So we have the following result from Theorem~\ref{th-1}.

\begin{corollary}\label{B-1}
The Bessel transformation $$Z_n(q)=\sum_{k=0}^n\frac{(n+k)!}{(n-k)!k!}X_k(q)$$ preserves the strong $q$-log-convexity.
\end{corollary}
Since the sequence $(\left(q/2\right)^k)_{k\geqslant0}$ is strongly $q$-log-convex, 
we can also present the following result from Corollary~\ref{B-1}.

\begin{corollary}[{\cite{cyc08}}]
The Bessel polynomials $y_n(q)$ form a strongly $q$-log-convex sequence.
\end{corollary}

\subsection{Strong $q$-log-convexity of the $r$-Dowling polynomials}
\hspace*{\parindent}
The $r$-Dowling polynomial $D_{m,r}(n,q)$~\cite{cj12} is the generating function of $r$-Whitney numbers of the second kind~\cite{im11},
which is a common generalization of the Whitney numbers of the second kind of the Dowling lattice and the $r$-Stirling numbers~\cite{Bro84}.
Cheon and Jung~\cite{cj12} established that the $r$-Dowling polynomials $D_{m,r}(n,q)$ have only negative real zeros, for all $r,m,n\geqslant1$.
In this subsection, we give the strong $q$-log-convexity of the $r$-Dowling polynomials using the following
relation established by Rahmani~\cite{mr14}

\begin{equation}\label{Dow-Bell}
D_{m,r}(n,q)=\sum_{k=0}^n\binom{n}{k}m^kr^{n-k}B_k\left(\frac{q}{m}\right),
\end{equation}
where $B_k(q)$ are the Bell polynomials.


From Theorem~\ref{th-1} and the recurrence relation~(\ref{binom}),
we also give the binomial transformation preserving the strong $q$-log-convexity directly,
which is first obtained by Zhu and Sun~\cite{ZS15}.

\begin{corollary}[{\cite{ZS15}}]\label{p-1}
The binomial transformation
$$Z_n(q)=\sum_{k=0}^n\binom{n}{k}X_k(q)$$ preserves the strong
$q$-log-convexity.
\end{corollary}

To prove the strong $q$-log-convexity of $r$-Dowling polynomials, 
we need two lemmas.

\begin{lemma}\label{l-3}
If $(x_n)_{n\geqslant 0}$ is a log-convex sequence and
$(X_n(q))_{n\geqslant 0}$ is a strongly $q$-log-convex sequence,
then $(x_nX_n(q))_{n\geqslant 0}$ is also a strongly
$q$-log-convex sequence.
\end{lemma}

\begin{proof}
Note that
\begin{eqnarray*}
& & x_{k+1}X_{k+1}(q)x_jX_j(q)-x_kX_k(q)x_{j+1}X_{j+1}(q)\\
&=&
(x_{k+1}x_j-x_kx_{j+1})X_{k+1}(q)X _j(q)+(X_{k+1}(q)X_j(q)-X_k(q)X_{j+1}(q))x_kx_{j+1}\\
&\geqslant_q&0.
\end{eqnarray*}
The inequality holds by the log-convexity of $(x_n)_{n\geqslant
0}$ and the strong $q$-log-convexity of $(X_n(q))_{n\geqslant 0}$.
Thus the sequence $(x_nX_n(q))_{n\geqslant 0}$ is strongly
$q$-log-convex.
\end{proof}

\begin{lemma}\label{c-1}
If $(X_n(q))$ is strongly $q$-log-convex,
then so does $(X_n(mq))_{n\geqslant 0}$ for $m\geqslant 1$.
\end{lemma}

\begin{proof}

Let $X_n(q)=\sum_{t=0}^{n}a(n,t)q^t$. 
Denote by $a(k,j;r,t)$ the following expression: 
$$a(k+1,t)a(j,r-t)+a(k+1,r-t)a(j,t)-a(k,t)a(j+1,r-t)-a(k,r-t)a(j+1,t).$$
By the assumption,
\begin{equation*}
X_{k+1}(q)X_j(q)-X_k(q)X_{j+1}(q)
=\sum_{r=0}^{k+s+1}\sum_{t=0}^ra(k,j;r,t)q^r\geqslant_q 0,
\end{equation*}
for $k\geqslant j\geqslant 0$.
Hence the coefficient $a(k,j;r,t)\geqslant 0$, for $k\geqslant j\geqslant 0$.

Thus we have
\begin{equation*}
X_{k+1}(mq)X_j(mq)-X_k(mq)X_{j+1}(mq)
= \sum_{r=0}^{k+s+1}\sum_{t=0}^ra(k,j;r,t)m^rq^r\geqslant_q 0,
\end{equation*}
i.e., $(X_n(mq))_{n\geqslant 0}$ is strongly $q$-log-convex, for $m\geqslant 1$.
\end{proof}

Now the strong $q$-log-convexity of the $r$-Dowling polynomials
follows immediately from Corollary~\ref{p-1} and the
relation~(\ref{Dow-Bell}).

\begin{corollary}\label{p-3}
The $r$-Dowling polynomials $D_{m,r}(n,q)$ form a strongly
$q$-log-convex sequence for $m\geqslant1$ and $r\geqslant0$.
\end{corollary}

\begin{remark}
An immediate consequence of Corollary~\ref{p-3} is the strong
$q$-log-convexity of the Dowling polynomials
$D_m(n,q)=D_{m,1}(n,q)$ (see~\cite{cyc08}).
\end{remark}
In particular, when $m=1$, 
we show the strong $q$-log-convexity of the $r$-Bell polynomials~\cite{im11}.

\begin{corollary}\label{p-4}
The $r$-Bell polynomials $B_{r}(n,q)$ form a strongly $q$-log-convex
sequence for $r\geqslant0$.
\end{corollary}

\begin{remark}
Note that the Bell polynomials $B_n(q)=B_0(n,q)$.
Hence the strong $q$-log-convexity of the Bell polynomials $B_n(q)$ follows from Corollary~\ref{p-4} (see~\cite{cyc08}).
\end{remark}













%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection*{Acknowledgements}
%Thanks to Professor Querty for suggesting the proof of
%Lemma~\ref{lem:Technical}.
The authors thank Prof. Y. Wu and Prof Y. Wang for their kind assistance in the preparation of this paper.
The authors thank the anonymous referee for his/her helpful suggestions
which greatly improved the original manuscript.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% \bibliographystyle{plain} 
% \bibliography{myBibFile} 
% If you use BibTeX to create a bibliography
% then copy and past the contents of your .bbl file into your .tex file



\end{document}